Author’s response to reviews

Title: Choroidal neovascularization in angioid streaks following microincision vitrectomy surgery: a case report

Authors:

Satoshi Katagiri (ktgr_two_ai@icloud.com)
Takaaki Hayashi (taka@jikei.ac.jp)
Hirotugu Takashina (two-shina@s7.dion.ne.jp)
Katsuya Mitooka (mitooka@jcom.aome.ne.jp)
Hiroshi Tsuneoka (tsuneh@jikei.ac.jp)

Version: 4 Date: 24 June 2013

Author’s response to reviews: see over
June 24, 2013

Prof. Andrew Dick
Section Editor of *BMC Ophthalmology*

MS: 4581721638792060

Dear Prof. Andrew Dick,

We would like to thank you and the reviewers for the constructive comments regarding our manuscript entitled "**Choroidal neovascularization after small-gauge vitrectomy in angioid streaks: a case report**". We have addressed each of the reviewers’ comments on a point-by-point basis. All changes were highlighted using red color text in the revised manuscript.

Sincerely,

Corresponding author
Satoshi Katagiri, MD
Department of Ophthalmology
The Jikei University School of Medicine
3-25-8 Nishi-shimbashi, Minato-ku, Tokyo 105-8461, Japan
Tel: +81-3-3433-1111 (ext. 3581); Fax: +81-3-3433-1936
E-mail: ktgr_two_ai@icloud.com
Reply to the reviewer (Dr. Winfried Amoaku)

1. The legend for the OCT in Fig 5 should indicate that 'there is significant but incomplete regression of CNV, and persistence of intra-retinal cysts'.

We agree the comment. We have changed to “Optical coherence tomography shows significant but incomplete regression of CNV, and persistence of intra-retinal cysts” in the text and the legend.

The correction has been made with red color fonts in the revised manuscript.

We wish to thank you for the helpful comments. We believe that our revised manuscript is now much improved.